Author's response to reviews

Title: Lung Gene Expression Changes in Streptozotocin-induced Diabetic Rats

Authors:

Erik van Lunteren (exv4@cwr.edu)
Michelle Moyer (michelle.moyer@va.gov)
Sarah Spiegler (spieglers@gmail.com)

Version: 2 Date: 9 October 2013

Author's response to reviews:

Revisions have been made as follows:

1. Name of ethics committee. This is now listed in the manuscript. (Department of Veterans Affairs (Veterans Health Administration))

2. Competing interests - This section is now included in the manuscript.

3. Authors' contributions - Authors' contributions are now included in the manuscript.

4. Deposition of microarray data. The information about the microarray data is now included in the manuscript (The data have been deposited in the NCBI Gene Expression Omnibus (GEO, http://www.ncbi.nlm.nih.gov/geo/) and are accessible through GEO Series accession number GSE15900.)